Author's response to reviews

Title: Antifungal wound penetration of amphotericin and voriconazole in combat-related injuries: case report

Authors:

Kevin S Akers (kevin.s.akers.mil@mail.mil)
Matthew P Rowan (matthew.p.rowan.vol@mail.mil)
Krista L Niece (krista.l.niece.ctr@mail.mil)
John C Graybill (john.c.graybill.mil@mail.mil)
Katrin Mende (katrin.mende.ctr@mail.mil)
Kevin K Chung (kevin.k.chung.mil@mail.mil)
Clinton K Murray (clinton.k.murray.mil@mail.mil)

Version: 2
Date: 31 October 2014

Author's response to reviews:

Dear BMC-Infectious Diseases Editorial Staff:

Thank you for your initial review of our submission. We have taken the requested comments into consideration and have made all required changes to the original version.

We have revised the original submission as follows:
- We have added line numbers and page numbers as requested.
- We have included the ethics statement in the Case Presentation section. This section was originally listed only in the Acknowledgements section and has now been added to the Methods/Case Presentation section as well.

Sincerely,

Kevin S. Akers, MD
United States Army Institute of Surgical Research
JBSA Ft. Sam Houston, TX
Kevin.s.akers.mil@mail.mil